Author's response to reviews

Title: Marketing depression care management to employers: design of a randomized controlled trial

Authors:

   Kathryn M Rost (kathryn.rost@med.fsu.edu)

Version: 3 Date: 22 February 2010

Author's response to reviews: see over
February 20, 2010

Dear Penny,

Thank you for your response to my inquiry. I am happy to accept the changes you have made to the manuscript.

With this cover letter please find attached the manuscript text with my own additions/deletions, which are highlighted in yellow as you suggested. These include 8 small word/phrase changes either to clarify the text or to reflect changed parameters of the study; the renumbering of reference; and the deletion of the words “the National Business Coalition on Health” from the acknowledgements. Please also find a revised Figure 1, changed only by the deletion of “with 30 employers each” in the first textbox, and a revised Figure 3, changed only by deleting superscript 5 and the note for it under the figure.

Thank you for your help on this project.

Sincerely,

Kathryn Rost